MATH 114: Elementary Calculus I

Instructor: Y. Lin, CAB 673, Tel: 492-7880 O.H. : MWF: 14:00-14:50pm or by appointment Time: MWF 11:00-11:50pm (D1) / 13:00-13:50pm (F1) Place CAB 239 TEXT : Calculus by James Stewart, 3rd Edition.

GRADING POLICY:

Final Exam: 50% Midterms: Oct. 23 and Nov. 27, 20% each Homework: 10%

CALCULATOR POLICY:

Calculators are not permitted on examinations.

HOMEWORK POLICY:

There will be 10 assignments during the term, each of equal weight. Assignments are due on Wednesdays ( Starting from Sept. 23). There is an assignment box on the 6th floor of CAB for submittion of assignments. Please make sure that you put your paper in the RIGHT BOX. Assignments may be turned in early, but can not be accepted late for any reasons. ONLY eight of your best assignments will be counted for your homework mark. Solutions to assignments will be posted in the glass covered bulletin board NO. 12 on the sixth floor of CAB opposite CAB 672 or on my office door after the assignments are turned in. www page will be setup soon for this course.

Outline:

Date Section from text Remarks Sept. 4-11 x1.1-1.3 Sept. 14-18 x1.3-2.1 Sept. 21-30 x2.1-2.5 Oct. 2-9 x2.5-2.8 Oct. 12-16 x2.8-2.10 Oct. 19-30 x3.1-3.2

First Midterm Oct. 23

Nov. 2-6 x3.3-3.5 Nov. 9-14 x3.5-3.7 Nov.16-20 x3.8-3.10 Nov. 23-30 x4.10-4.2

Second Midterm Nov. 27

Dec. 1-4 x4.3-4.5 Dec. 7-9 x4.5–review Final Exam